Author’s response to reviews

Title: Examining the ethical and social issues of health technology design through the public appraisal of prospective scenarios: A study protocol describing a multimedia-based deliberative method

Authors:

Pascale Lehoux (pascale.lehoux@umontreal.ca)
Philippe Gauthier (philippe.gauthier.2@umontreal.ca)
Bryn Williams-Jones (bryn.williams-jones@umontreal.ca)
Fiona A Miller (fiona.miller@utoronto.ca)
Jennifer J Fishman (jennifer.fishman@mcgill.ca)
Myriam Hivon (myriam.hivon@umontreal.ca)
Patrick Vachon (p.vachon@umontreal.ca)

Version: 3
Date: 22 May 2014

Author’s response to reviews:

Dear Paul:

Many thanks for inviting us to revise our study protocol. This is greatly appreciated.

Following your suggestion, we have clarified which steps had been completed and which steps remain to be done. This clarification involved: 1) revising the abstract; 2) reformulating the Study Status section; and 3) adding a note to Figure 1 that reiterates the study timeline.

We hope these changes will meet your expectations.

Best wishes,